Author's response to reviews

Title: Large cell neuroendocrine carcinoma of the lung presenting as pseudo achalasia: a case report

Authors: 

Ahmad Anaizi (ahmad.anaizi7@gmail.com)  
Amna Rizvi-Toner (amna-rizvi@uiowa.edu)  
Ron Schey (ron.schey@tuhs.temple.edu)

Version: 4 Date: 13 October 2014

Author's response to reviews: see over
Author’s response to reviews

A case of Metastatic esophageal lesion presenting as a Paraneoplastic Syndrome

Authors:

Ahmad Anaizi (ahmad-anaizi@uiowa.edu)
Amna S Rizvi-Toner (amna-rizvi@uiowa.edu)
Ron Schey (ron.schey@tuhs.temple.edu)

Author's response to reviews:

1. Title change: Done, “Large cell neuroendocrine carcinoma of the lung presenting with pseudo achalasia”

2. The case is not a primary esophageal net, therefore no need to mention about esophageal net in the text: Agree, and deleted

3. The text should be changed as the case initially thought to be paraneoplastic, however in the long term follow up metastasis to the esophageal was shown: Agree and text adjusted